Title: A Mid Year Comparison Study of Career Satisfaction and Emotional States between Residents and Faculty at One Academic Medical Center

Authors:

Donald E Girard MD (girardd@ohsu.edu)
Dongseok Choi PhD (choi@ohsu.edu)
Jamie Dickey PhD (dikeyja@ohsu.edu)
Kristen Wessel PhD (wesselkr@ohsu.edu)
Donald Austin MD, MPH (austind@ohsu.edu)

Version: 3 Date: 30 June 2006

Author's response to reviews: see over
30 June 2006

Editors
BMC Medical Education

Dear Editors:

Thank you very much for accepting our manuscript in a timely manner. We believe that we have addressed all of the comments and formatting issues. In reviewing the final version of the manuscript we took the opportunity to add a comma on page 5, amend the wording of the first sentence of the Results Section on page 6, and make a very minor change in the wording at the end of the first paragraph on page 8. We hope these minor changes are acceptable. Again, we appreciate your timely and positive review and look forward to the article’s publication. If there are any further changes, please notify me.

Most sincerely,

Donald E. Girard, MD
Professor of Medicine
Associate Dean,
Graduate and Continuing Medical Education
Oregon Health & Science University
Portland, Oregon 97239